\documentstyle[12pt]{article}
%\documentstyle[12pt]{prop}
\headheight 40pt
\headsep 30pt
\textwidth 6.5in
\textheight 8.7in
\oddsidemargin 0pt
\topmargin -45pt
\renewcommand{\baselinestretch}{1.7}
\pagestyle{empty}
\def\bib{\parskip=0pt\par\noindent\hangindent\parindent
    \parskip =2ex plus .5ex minus .1ex}
\begin{document}

% Include Rokicki's epsf.sty file for Encapsulated PostScript graphics
\input{epsf.sty}

\vskip6truept
\begin{center}
{\bf ASTR 5720 Galaxies \& Cosmology Fall 1997.  Problem Set 1.\\
Due Thur 4 Sept}
\end{center}
\vskip4truept

Part of the idea of this problem set is to allow you to investigate what's
out there on the web in cosmology.
Be prepared to defend your choice in class on Thur 4 Sept.

\begin{enumerate}
\item
Give me a title for your web project.
Outline briefly what you plan to do,
and what you find appealing about your proposal.
Describe what web sites you may have found which may prove useful
to your project
(URL please, if I don't already have it).
\end{enumerate}

\end{document}
